Author’s response to reviews

Title: Understanding why adult participants at the World Senior Games Choose a Healthy Diet

Authors:

Prof Ray M Merrill (Ray_Merrill@byu.edu)  
Eric C Shields (ecs26@email.byu.edu)

Version: 2  Date: 6 Nov 2003

Dear Ray,

There are a few formatting changes to be made before your manuscript is suitable for publication.

Formatting changes requested

* Title- Please remove all unnecessary capitalisation from the title
* Country - Please remember to include country in the affiliations' details.
* Corresponding author - The corresponding author should be clearly indicated on the title page

Sincerely,

Nehme Gebrayel, MD

These changes were made and reflected in the revised manuscript.